% Example Document for ``IEEEUltraSymp'' package for 
% 2002 IEEE Ultrasonics Symposium Proceedings papers
%
% Written by Ronald Kumon based on the notes by Michael Shell 
% in ``conf_notes.txt'' of the ``IEEEtran'' distribution (available at 
% http://www.ieee.org/organizations/pubs/transactions/stylesheets.htm)
% 
% Requires:  
% 1. IEEEtran_v15.cls : IEEE Transactions LaTeX2e class file, 
% 2. ieeeultrasymp.sty: IEEE Ultrasonics Symposium style file
% Optional:  
% 1. graphicx.sty: Allows for input of Postscript figures 
% 2. IEEEbib.bst : Automatically formats references from BibTeX 
%                  databases to IEEE bibliographic btyle
%
% Test Platform: Red Hat Linux 7.0 i386 system with teTeX 1.0.7.
%
% To convert to Postscript from DVI:
% dvips -Ppdf -G0 -o ultra-example.ps ultra-example.dvi
% Ppdf:  includes printer options appropriate for PDF conversion
%        including font embedding
% G0  :  character remapping such that ligatures are not included
%        for Times-Roman font.
%
% To convert from Postscript to PDF using PStill (www.pstill.com):
% pstill -ccccgipt -o ultra-example.pdf ultra-example.ps
% c:  compression (more c's give more compression; up to four)
% g:  take size from Postscript file
% p:  include all fonts as partial fonts
% i:  include all non-standard fonts
% t:  map graphics transfer functions from PostScript to PDF. 
% (PDF conversion can be accomplished in other ways as well, e.g.,
% by using recent versions of Ghostscript.)
%
%**********************************************************************
% Legal Notice:
% This code and advice is offered as-is without any warranty either
% expressed or implied; without even the implied warranty of
% MERCHANTABILITY or FITNESS FOR A PARTICULAR PURPOSE!
% User assumes all risk.
% In no event shall IEEE or any contributor to this code
% be liable for any damages or losses, including, but not limited to,
% incidental, consequential, or any other damages, resulting from the
% use or misuse of any information contained here.
% 
% All statements made here are the opinions of their respective
% authors and are not necessarily endorsed by the IEEE.
%
% This code/text is in the public domain,   
% and may be freely used, distributed and modified.
% Please retain the contribution notices and credits.
%**********************************************************************

% use 11pt, twocolumn, single sided
\documentclass[11pt,twocolumn]{IEEEtran}
% use 11pt, onecolumn, single sided, draft mode
%\documentclass[11pt,onecolumn,draftcls]{IEEEtran}
\usepackage{ieeeultrasymp}
\usepackage{graphicx}

% Fill in the appropriate path here, if Postscript files are not 
% in the same directory as this file.
%\graphicspath{{}}

\begin{document}
% Set these offsets if your printer doesn't output quite on center
%\setlength{\voffset}{0.1875in}
%\setlength{\hoffset}{-0.1875in}

% Now, force the title to be in 14 point small caps:
\title{
% Adds required extra 3/8'' on first page.  Note that the space
% needs to be put in the title, otherwise this command will not work.
\vspace{0.375in}
\scshape\large Demonstration of Nuclear Fusion in an Ordinary Clay
Flower Pot}

% IEEEtran normally already uses \large (12pt) for author's names.
% However, we must add \normalsize for the affiliation and use line
% separators as desired.
% Note: V1.5 and later support the use of the \and operator to list multiple
% authors of different affiliations. Within the IEEEtran.cls file, this is
% done by enclosing the \author{} text within a tabular environment - just
% like article.cls does. 
% 
\author{
Albert Einstein\textsuperscript{*} 
and Alexander Bell\textsuperscript{\dag}\\
Princeton University, 
777 Luckey Street, 
Princeton, NJ 90210--1234, USA
\and
Richard Feynman\textsuperscript{\ddag}\\
California Institute of Technology, 
111 Main Street, Pasadena, CA 99999-9999, USA}

% Make title page, as usual
\maketitle 

% Enter abstract
\begin{abstract}
This work demonstrates a sustainable nuclear fusion reaction of hydrogen 
using a clay flower port as a reactor vessel. Our novel approach uses
a ``charge mirror" that reduces the electromagnetic repulsion between 
nuclei enough to allow fusion initiation at room temperature.
The device can also be used as a secure error-free transgalactic communications
pipe with zero latency and near infinite bandwidth.   
\end{abstract}

% Begin paper
\section{\label{sec:intro}Introduction}
This work demonstrates a sustainable nuclear fusion reaction of hydrogen 
using a clay flower port as a reactor vessel. Our novel approach uses
a ``charge mirror" that reduces the electromagnetic repulsion between 
nuclei enough to allow fusion initiation at room temperature.
The device can also be used as a secure error-free transgalactic communications
pipe with zero latency and near infinite bandwidth.   

\subsection{\label{sec:intro:further}Further Introduction}
This work is completely radical.  We really don't need to cite any 
references.  However, we will cite ourselves~\cite{epr35,feyn54,feynman59} 
just to increase our citation record.  Here are some 
more~\cite{einstein67,bell24,bell34}.

\subsubsection{\label{sec:intro:further:more}More Introduction}
This work demonstrates a sustainable nuclear fusion reaction of hydrogen 
using a clay flower port as a reactor vessel. Our novel approach uses
a ``charge mirror" that reduces the electromagnetic repulsion between 
nuclei enough to allow fusion initiation at room temperature.
The device can also be used as a secure error-free transgalactic communications
pipe with zero latency and near infinite bandwidth.   

\subsubsection{\label{sec:intro:further:evenmore}Even More Introduction}
This work demonstrates a sustainable nuclear fusion reaction of hydrogen 
using a clay flower port as a reactor vessel. Our novel approach uses
a ``charge mirror" that reduces the electromagnetic repulsion between 
nuclei enough to allow fusion initiation at room temperature.
The device can also be used as a secure error-free transgalactic communications
pipe with zero latency and near infinite bandwidth.   

\subsection{\label{sec:intro:ending}Ending Introduction}
This work demonstrates a sustainable nuclear fusion reaction of hydrogen 
using a clay flower port as a reactor vessel. Our novel approach uses
a ``charge mirror" that reduces the electromagnetic repulsion between 
nuclei enough to allow fusion initiation at room temperature.
The device can also be used as a secure error-free transgalactic communications
pipe with zero latency and near infinite bandwidth.   

This work demonstrates a sustainable nuclear fusion reaction of hydrogen 
using a clay flower port as a reactor vessel. Our novel approach uses
a ``charge mirror" that reduces the electromagnetic repulsion between 
nuclei enough to allow fusion initiation at room temperature.
The device can also be used as a secure error-free transgalactic communications
pipe with zero latency and near infinite bandwidth.   

See Sec.~\ref{sec:intro}, Sec.~\ref{sec:intro:further}, and 
Sec.~\ref{sec:intro:further:evenmore} for details.

\section{\label{sec:another}Another section}
This work demonstrates a sustainable nuclear fusion reaction of hydrogen 
using a clay flower port as a reactor vessel. Our novel approach uses
a ``charge mirror" that reduces the electromagnetic repulsion between 
nuclei enough to allow fusion initiation at room temperature.
The device can also be used as a secure error-free transgalactic communications
pipe with zero latency and near infinite bandwidth.   

Figure~\ref{fig:claypot} shows the experimental apparatus.
The center asterisk shows the location of the fusion reaction.

\begin{figure}
\begin{center}
\includegraphics*[width=3.3in]{figure.eps}
\caption{\label{fig:claypot}
Schematic diagram of the experimental apparatus 
for nuclear fusion in a clay flower pot.}
\end{center}
\end{figure}

This work demonstrates a sustainable nuclear fusion reaction of hydrogen 
using a clay flower port as a reactor vessel. Our novel approach uses
a ``charge mirror" that reduces the electromagnetic repulsion between 
nuclei enough to allow fusion initiation at room temperature.
The device can also be used as a secure error-free transgalactic communications
pipe with zero latency and near infinite bandwidth.   

% Extend page length to balance the columns.  (See note below.)
% Note that argument for your paper will probably be different
\enlargethispage{-0.50in}

\section{\label{sec:theory}Theory}
The theory can't possibly be understood by anyone but us.  
Nevertheless, we give here the key equation
\begin{equation}
\label{eqn:energy}
E^2=(pc)^2+(mc^2)^2\,.
\end{equation}
Obviously, Eq.~(\ref{eqn:energy}) says it all.

This work demonstrates a sustainable nuclear fusion reaction of hydrogen 
using a clay flower port as a reactor vessel. Our novel approach uses
a ``charge mirror" that reduces the electromagnetic repulsion between 
nuclei enough to allow fusion initiation at room temperature.
The device can also be used as a secure error-free transgalactic communications
pipe with zero latency and near infinite bandwidth.   

This work demonstrates a sustainable nuclear fusion reaction of hydrogen 
using a clay flower port as a reactor vessel. Our novel approach uses
a ``charge mirror" that reduces the electromagnetic repulsion between 
nuclei enough to allow fusion initiation at room temperature.
The device can also be used as a secure error-free transgalactic communications
pipe with zero latency and near infinite bandwidth.   

This work demonstrates a sustainable nuclear fusion reaction of hydrogen 
using a clay flower port as a reactor vessel. Our novel approach uses
a ``charge mirror" that reduces the electromagnetic repulsion between 
nuclei enough to allow fusion initiation at room temperature.
The device can also be used as a secure error-free transgalactic communications
pipe with zero latency and near infinite bandwidth.   

\section{\label{sec:conclusion}Conclusion}
This work demonstrates a sustainable nuclear fusion reaction of hydrogen 
using a clay flower port as a reactor vessel. Our novel approach uses
a ``charge mirror" that reduces the electromagnetic repulsion between 
nuclei enough to allow fusion initiation at room temperature.
The device can also be used as a secure error-free transgalactic communications
pipe with zero latency and near infinite bandwidth.   

% References
% Note that the number section heading is generated automatically
% by the \bibliography command.  If you put your references in yourself, 
% then you will have to use the commands:
% \section{References}
% \begin{thebibliography}{99}
% \bibitem{...}
% \end{thebibliography}
\bibliography{ieeesamp}
\bibliographystyle{IEEEbib}

% Add email addresses here, if desired.
\vspace{\baselineskip}
\noindent
\textsuperscript{*}Email: einstein@princeton.edu\\
\textsuperscript{\dag}Email: bell@lucent.com\\
\textsuperscript{\ddag}Email: feynman@caltech.edu

% From conf_notes.txt in the ieeetran distribution:
% ``Now somewhere just after the beginning of the first column on the last
% page, do a:
%   \enlargethispage{-X.Xin}                                    
% where X.X is chosen to shorten the first column so that both 
% columns have the same amount of text.
% This is a really crude hack, but it does the job.
% If your last page has nothing but references (which began on the
% previous page), you may have to position this command between references.
% This means you will have to manually copy in your .bbl file if you are
% relying on bibtext (via \bibliography).
% Needless to say, X.X is highly dependent on document content. So, 
% \enlargethispage{-X.Xin} only after you are sure your document is in its 
% final form.
%   
% You can also try the flushend.sty package, 
% (available on CTAN or http://www.vtex.lt/tex/download/macros/ ).
% However, the version I have (1997/10/01), created a little spacing
% anomaly between two references in the second column of my paper. So,
% I went back to the manual way above of doing things.''
        
\end{document}       
